Author’s response to reviews

Title: The Impact of Cathelicidin, the Human Antimicrobial Peptide LL-37 in Urinary Tract Infections

Authors:

Ibrahim Babikir (almakibrahim@hotmail.com)
Elsir Abugroun (elsir.abugroun@yahoo.com)
NaserEldin Bilal (nasereldin2001@gmail.com)
Abdullah Alghasham (ghasham@qumed.edu.sa)
Elmuataz Abdalla (almuataz1960@yahoo.co.uk)
Ishag Adam (ishagadam@hotmail.com)

Version: 4 Date: 03 Dec 2017

Author’s response to reviews:

Point-by-point response 03 Dec. 2017

Dear Editor,

Thanks again for your kind efforts to help us to improve the manuscript to a better one.

We are so grateful for the helpful comments/corrections. Below are the responses to the editor comments/corrections.

Author’s response:

Editor Comments/reports:

1. Please reformat your abstract so that it follows the following format:

   Background: the context and purpose of the study

   Methods: how the study was performed and statistical tests used

   Results: the main findings

   Conclusions: brief summary and potential implications
Response

Yes, it has been done. Please see the abstract.